Author’s response to reviews

Title: Online Cost-Effectiveness ANalysis (OCEAN): A user-friendly interface to conduct cost-effectiveness analyses for cervical cancer

Authors:

David Moriña (dmorina@mat.uab.cat)
Jose Ignacio Martí-Gil (jimartigil@gmail.com)
Pedro Puig (ppuig@mat.uab.cat)
Mireia Diaz (mireia@iconcologia.net)

Version: 5 Date: 23 Aug 2020

Author’s response to reviews:

Editor Comments:

1. Authors' contributions

– We have noted that author [PP] is missing in the listed authors' contributions. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

PP contributions to the paper have been added.

2. Availability of data and materials

– The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based.

The “Availability of data and materials” section has been rewritten to indicate that the example used in the paper can be reproduced by using sample input files that can be downloaded from the OCEAN tool.

3. Consent for publication

– Please amend your “Consent for publication” section in your Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical
details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

The section “Consent for publication” has been added following the Editor’s advice.

4. Clean copy

– At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

A clean copy of the final version of the manuscript has been uploaded. Figures have been uploaded as separate files and are cited within the text.